Author's response to reviews

Title: Association of socioeconomic and behavioral factors with adult mortality: Analysis of data from verbal autopsy in Addis Ababa, Ethiopia.

Authors:

Awoke Misganaw (asterawoke2007@gmail.com)
Damen Haile Mariam (damen_h@hotmail.com)
Tekebash Araya (tekebash@gmail.com)

Version: 8 Date: 11 June 2013

Author's response to reviews: see over
Authors' response to reviews

Title: Association of socioeconomic and behavioral factors with adult mortality: Analysis of data from verbal autopsy in Addis Ababa, Ethiopia.

Version: 7 Date: 31 May 2013

Reviewer: Jayadeep Patra

The authors would like to thank the reviewer for his/her constructive comments. We have given our response to the comments one by one as follows and incorporated in the text document.

Following are our response to comments for authors:

1. Accept the paper with one more round of revision in language check

   **Response: A native speaker has edited the language**

2. Presentation of tables is NOT professionally done - they are redundant. It should adhere to journal style

   **Response: Presentation of tables has been modified according to the journal style.**

3. Relevance to the topic is OK